Author's response to reviews

Title: Polymorphisms cMyc-N11S and p27-V109G and breast cancer risk and prognosis

Authors:

Jane C Figueiredo (jane@mshri.on.ca)
Julia A Knight (knight@mshri.on.ca)
Stewart Cho (cho@mshri.on.ca)
Sevtap Savas (savas@mshri.on.ca)
Venus Onay (onay@mshri.on.ca)
Laurent Briollais (laurent@mshri.on.ca)
Pamela J. Goodwin (pgoodwin@mtsinai.on.ca)
John R. McLaughlin (john.mclaughlin@cancercare.on.ca)
Irene L. Andrilis (andrilis@mshri.on.ca)
Hilmi Ozcelik (ozcelik@mshri.on.ca)

Version: 5 Date: 8 June 2007

Author's response to reviews:

Jane Figueiredo, PhD
Samuel Lunenfeld Research Institute
Mount Sinai Hospital
60 Murray Street Box 18
Toronto, ON M5T 3L9 CANADA
jane@mshri.on.ca

RE: 572327718122179

Dear Dr. Edmunds:

The final changes have been made: (i) include a sentence indicating all authors read and approved the final manuscript (page 14) and (ii) the list of authors in the manuscript written exactly as they are in the submission system (page 1).

Thank you,

Jane Figueiredo